Author’s response to reviews

Title: Experience of maintaining tooth brushing for children born with a cleft lip and/or palate

Authors:
Yin-Ling Lin (yin-ling.lin@manchester.ac.uk)
Karen Davies (karen.davies-3@manchester.ac.uk)
Peter Callery (peter.callery@manchester.ac.uk)

Version: 2  Date: 17 Oct 2016

Author's response to reviews:
Dear reviewers,

We are extremely grateful for your time and constructive comments. We have revised the article and responded to the comments, this can be found in the attached document entitled, 'Response to reviewers'. In this document, we have identified the line numbers and the pages where the changes could be found in our response. Moreover, we have left the track changes in the manuscript so that you can see what changes have been made.

We are looking forward to hearing back from you.

Many thanks,

Yin-Ling